Author's response to reviews

Title: Perinatal outcomes in a South Asian setting with high rates of low birth weight

Authors:

Kuryan George (kuryan@cmcvellore.ac.in)
Jasmin Prasad (jasminep@cmcvellore.ac.in)
Daisy Singh (pushp@cmcvellore.ac.in)
Shantidani Minz (shantidani@cmcvellore.ac.in)
David S Albert (david.selvapandian@gmail.com)
Jayaprakash Muliyil (jayaprakash@cmcvellore.ac.in)
K S Joseph (ksjoseph@dal.ca)
Jyothi Jayaraman (j.jayaraman@ns.sympatico.ca)
Michael S Kramer (michael.kramer@mcgill.ca)

Version: 4 Date: 5 February 2009

Author's response to reviews:

1. Author groups - Please remove the group names (Christian Medical College and Canada Collaborative Group) and "The Group members include" from the title page. You should then add a note in the Acknowledgements and Authors' contributions defining the members contained within the group, and the role of the group within this study.

This has been done.

2. Conclusions - Please insert a Conclusions section after the Discussion. We recommend including a Conclusions heading above the paragraph which begins 'In conclusion...'

Done.

3. Table size - Tables must be A4 portrait (maximum size 176mm x 230mm). If you cannot re-orientate ......

Tables have been reformatted to A4 portrait size.

4. References 38 and 41, In press - Have these articles now been published? If so, please provide the citation details.

References 37 and 40 have been published and hence the citations have been updated.

5. Introduction - Please rename this section 'Background'.

Done.
6. References - Please move these to after the Acknowledgements.
Done.

7. Figure legends - Please move these to before the Tables.
Done.

8. Figure cropping
Done.